**Author's response to reviews**

**Title:** Identifying Inaccuracies on Emergency Medicine Residency Applications

**Authors:**

Eric D Katz (KATZE@MSNOTES.WUSTL.EDU)
Lee Shockley (lee.shockley@dhha.org)
Lawrence Kass (lkass@hmc.psu.edu)
David Howes (dhowes@medicinebsd.uchicago.edu)
Janis P Tupesis (Janis_tupesis@hotmail.com)
Christopher Weaver (chswave@iupui.edu)
Osman R Sayan (ors1@columbia.edu)
Jason Begue (jbegue@uabmc.edu)
Diamond Vrocher (dvrocher@uabmc.edu)
Jackie Frazer (jfrazier@uabmc.edu)
Timothy Evans (tcevans@hsc.vcu.edu)
H GENE Hern (gene_hern@yahoo.com)
Ralph Riviello (Ralph.riviello@jefferson.edu)
Antonio Rivera (riveraa@msnotes.wustl.edu)
Keith Kinoshita (kinoshik@msnotes.wustl.edu)
Edward Ferguson (fergused@msnotes.wustl.edu)

**Version:** 5 **Date:** 16 August 2005

**Author's response to reviews:**

To whom it may concern,

1. Title and abstract made identical on submission and manuscript
2. Affiliations corrected as described. Links made by superscripted reference, with country listed.
3. Keywords line removed
4. Word count line removed
5. References changed from italicized superscript to brackets.
6. Section heading, table titles and title punctuation corrected.